Author's response to reviews

Title: The knowledge and attitudes about breast cancer of women in a rural area of western region in Turkey

Authors:

- Pinar Erbay Dundar (pinarerbaydundar@yahoo.com)
- Dilek Ozmen (ozmendilek@yahoo.com)
- Beyhan Ozyurt (beyhanozyurt@hotmail.com)
- Gokce Haspolat (gokcemmdr@hotmail.com)
- Filiz Akyildiz (dr.filizakyildiz@hotmail.com)
- Sumeyra Coban (dr.sumeyra@hotmail.com)
- Gamze Cakiroglu (drgamzecakoglu@hotmail.com)

Version: 2 Date: 14 November 2005

Author's response to reviews:

Authors' contribution section is added after the competing interest section, references are cited in the text using consecutive numbers in square brackets and the questionnaire is provided.